Author's response to reviews

Title: Effects of the El Nino-Southern Oscillation on dengue epidemics in Thailand, 1996-2005

Authors:

Mathuros Tipayamongkholgul (mathurost@hotmail.com)
Chi-Tai Fang (fangct@ntu.edu.tw)
Suratsawadee Klinchan (noosurats@hotmail.com)
Chung-Ming Liu (liucm@ntu.edu.tw)
Chwan-Chuen King (chwanchuen@gmail.com)

Version: 3 Date: 14 November 2009

Author's response to reviews: see over
November 14, 2009

RE: BMC Public Health MS 949588073264443

Dear Dr. Norton:

Thank you so much for accepting our work for publication in your prestigious journal. We have adjusted the format of our manuscript, as required by the BioMed Central Editorial Production Team:

1. We adjusted the labeling of Figures and Tables to the following order:
   
   Figure 1 (Figure 1 in the previous version); Figure 2 (Figure S1 in the previous version);
   Figure 3 (Figure 2 in the previous version); Figure 4 (Figure 3 in the previous version),
   and Figure 5 (Figure 4 in the previous version);
   
   Table 1 (Table 1 in the previous version); Table 2 (Table 2 in the previous version);
   Table 3 (Table 3 in the previous version); Table 4 (Table 4 in the previous version);
   Table 5 (Table S1 in the previous version); Table 6 (Table 5 in the previous version);
   Table 7 (Table 6 in the previous version); and Table 8 (Table S2 in the previous version)

2. In this revised version, we updated the hyperlink of reference # 2 to the following web site: [http://www.searo.who.int/EN/Section10/Section332_1101.htm](http://www.searo.who.int/EN/Section10/Section332_1101.htm), since the previous hyperlink: [http://www.searo.who.int/EN/Section10/](http://www.searo.who.int/EN/Section10/) is no longer functioning.

3. In this revised version, the underlines and the line numbers were removed.

4. In this revised version, the appendix was placed before the acknowledgement section.

5. In this revised version, the size of Table 6 was reduced to two pages by decreasing the line spacing to 1.

6. The image files of Figures 1–5 were of 3,508x2,480 pixels, which will provide a 300 dpi resolution in the A4 size. Please tell me if a different resolution is required.

Thank you again for your kind consideration.

Sincerely yours,
Chi-Tai Fang, MD, PhD  
Associate Professor  
Graduate Institute of Epidemiology  
College of Public Health, National Taiwan University  
17 Xu-Zhou Road, Taipei 100, Taiwan  
Phone: +886-2-3366-8035  
Fax: +886-2-2395-2133  
E-mail: fangct@ntu.edu.tw